\begin{table}%T2
   \caption{Objects with photometry suggesting they are L
   dwarfs. Our photometric spectral classifications are based on the
   colours in Kirkpatrick et~al. (\cite{Kirkpatrick00}). }
         \label{Redtab1}
%\centerline
 {\small
\begin{tabular}{lcccccccc}
\hline\hline
Name&{$\mu$}&{$I$}&{$J$}&{$H$}&{$K_{\rm s}$}&Photometric&Previous&Previously\\
&(arcsecs/yr)&&&&&Spectral Class&Spectral Class&found in\\
\hline
SIPS1753--6559&0.393&17.53&14.10&13.11&12.42&L2.5&&\\
SIPS2045--6332&0.223&15.95&12.62&11.81&11.21&L0&&\\
SIPS2255--5713&0.341&17.85&14.08&13.19&12.58&L1.5&L5.5&$^1$\\
SIPS0128--5545&0.293&17.45&13.78&12.92&12.34&L1&L1&$^1$\\
SIPS1341--3052&0.196&17.96&14.61&13.73&13.08&L2&&\\
SIPS2026--2943&0.417&17.78&14.80&13.95&13.36&L1&&$^7$\\
SIPS0316--2848&0.206&17.88&14.58&13.77&13.11&L1.5&L0&$^2$\\
SIPS0614--2019&0.308&17.90&14.78&13.90&13.38&L0.5&&\\
SIPS1058--1548&0.249&17.61&14.16&13.23&12.53&L2.5&L3$^4$&$^3$\\
SIPS1228--1547&0.415&17.62&14.38&13.35&12.77&L2&L5$^4$&$^3$\\
SIPS0847--1532&0.246&16.42&13.51&12.63&12.06&L1&L2&$^2$\\
SIPS0408--1450&0.256&17.44&14.22&13.34&12.82&L1&L2&$^2$\\
SIPS0058--0651&0.304&17.67&14.31&13.44&12.90&L1&L0$^2$&$^5$\\
SIPS0539--0059&0.363&17.84&14.03&13.10&12.53&L1.5&L5&$^6$\\
\hline
\end{tabular}}
\medskip
\par
Citation key: $^1$ Kendall et~al. (\cite{Kendall06}), $^2$ Cruz et~al. (\cite{Cruz03}), $^3$ Delfosse et al. (\cite{Delfosse99}), $^4$ Kirkpatrick et~al. (\cite{Kirkpatrick99}), $^5$ Dahn et~al. (\cite{Dahn02}), $^6$ Fan et~al. (\cite{Fan00}), $^7$ Cruz et~al. (\cite{Cruz07}).
\end{table}